Author’s response to reviews

Title: Complete resection of a giant mediastinal teratoma occupying the entire right hemithorax in a 14-year-old boy

Authors:

Honglin Zhao (zhaohonglintj@163.com)
Daxing Zhu (zhudaxingtj@163.com)
Qinghua Zhou (cnliuqian@gmail.com)

Version: 2 Date: 8 April 2014

Author’s response to reviews:

Dear Editor,

This study was approved the ethical committee of Tianjin Medical University General Hospital. The Competing Interests, Authors’ Contributions are added in the article.

Thank you.

kind regards
Qinghua Zhou M.D.